Author's response to reviews

Title: Keratitis by Fusarium temperatum, a novel opportunist

Authors:

Abdullah M.S Al-Hatmi (a.alhatmi@cbs.knaw.nl)
Alexandro Bonifaz (a.bonifaz@yahoo.com.mx)
G. Sybren de Hoog (s.hoog@cbs.knaw.nl)
Leticia Vazquez-Maya (letivaz@yahoo.com.mx)
Karla Garcia-Carmona (karlapaola10@gmail.com)
Jacques F. Meis (j.meis@cwz.nl)
Anne D. van Diepeningen (a.diepeningen@cbs.knaw.nl)

Version: 8  Date: 23 October 2014

Author's response to reviews: see over
Dear Editor,

Thank you very much for forwarding further additional modification for revisions concerning our manuscript "Keratitis by Fusarium temperatum, a novel opportunist" (#MS: 1930636285131411). Kindly note that all the editorial comments were considered and the manuscript has been modified:

Response to Editorial Revisions:

1. The paper contains age, sex and location of the farmer. These individual information/data compromises anonymity. We would like to confirm whether written informed consent was obtained from the patients to publish their data and information. If not, please ensure that the age, sex and location of the farmer is removed from the paper to maintain confidentiality.

Response: In fact, we have tried to reach the patient again to sign consent to publish his data within the last couple of days but unfortunately we could not so we decided to remove the age, sex and the location of the farmer from the paper and replaced them with agriculture worker.

2. Please also clarify whether the all procedures carried out in this study were done as part of standard patient care.

Response: All the procedures used are standard routine procedures that are regularly employed to diagnose and manage patients with fungal keratitis at the Hospital General de México as part of standard patient care.

3. Research involving human subjects, human material, or human data, must have been performed in accordance with the Declaration of Helsinki and must have been approved by an appropriate ethics committee. A statement detailing this, including the name of the ethics committee and the reference number where appropriate, must appear in all manuscripts reporting such research.

Response: The patient is part of a study that has the authority of a scientific and ethics committee approved this study and the following statement was added to the paper under the Methods section.

(The study protocol was approved by the Scientific and Ethics Committees of the Hospital General de México (approval number DIC/12/102/3/23) and was performed in accordance with the ethical principles described in the 1964 Declaration of Helsinki. Informed written consent was obtained from all participants prior to their inclusion in the study).
4. Please check the data deposition links provided as an error message appears when attempting to access the treeBase link.

Response: We have tried to rectify the TreeBase website regarding this issue; they sent us the following link which can be opened from any browsers. We have tried the following link several times and it also works. Citation for study: 16415.

http://purl.org/phylo/treebase/phylows/study/TB2:S16415?x-access-code=ac517f778e8736faf6a6a3e9e4f1b263&format=html

The manuscript is formatted according to BMC guidelines using provided link.

Looking forward to hearing from you soon.

Kind regards,

Abdullah Al-Hatmi
Corresponding author for this manuscript